Author's response to reviews

Title: Prognostic value analysis of urokinase-type plasminogen activator receptor in oral squamous cell carcinoma: an immunohistochemical study

Authors:

  Roberta Bacchiocchi (bacchiocchi@univpm.it)
  Corrado Rubini (c.rubini@univpm.it)
  Elisa Pierpaoli (elisa.pierpaoli@gmail.com)
  Giulia Borghetti (giuliaborghetti@hotmail.it)
  Pasquale Procacci (pasqualeprocacci@hotmail.com)
  Pier Francesco Nocini (pierfrancesco.nocini@univr.it)
  Andrea Santarelli (andreasantarelli@tele2.it)
  Romina Rocchetti (romina.rochetti@libero.it)
  Domenico Ciavarella (drdomenicociavarella2@yahoo.it)
  Lorenzo Lo Muzio (llomuzio@tin.it)
  Francesca Fazioli (fazioli@univpm.it)

Version: 4 Date: 1 August 2008

Author's response to reviews:

  Dr Robin Cassady-Cain
  Assistant Editor
  BMC-series Journals
  1th August 2008
  MS: 2007313741819333

Dear Dr Cassady-Cain,

Thank you for your mail of July 31, 2008 regarding our manuscript “Prognostic value analysis of urokinase-type plasminogen activator receptor in oral squamous cell carcinoma: an immunohistochemical study” (manuscript 2007313741819333).

Following are the formatting changes performed.
1) References list has been updated conforming to final manuscript version.
2) Running head has been removed.
3) Tables have been renamed using standard numbers.

Your sincerely,

Dr Roberta Bacchiocchi